

# **Geophysical Research Letters**<sup>•</sup>

# **RESEARCH LETTER**

10.1029/2021GL097350

# **Special Section:**

Advancing prediction of coastal marine ecosystems

#### **Key Points:**

- A large-scale precursor to Northeast Pacific marine heatwaves (MHWs) with anomalies in both the North and Tropical Pacific is identified
- This optimal precursor includes initial sea surface height anomalies that are key to MHW growth
- The initial condition leading to optimal development of Northeast Pacific MHWs can also lead to subsequent Central Pacific El Niño events

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

A. Capotondi, antonietta.capotondi@noaa.gov

#### Citation:

Capotondi, A., Newman, M., Xu, T., & Di Lorenzo, E. (2022). An optimal precursor of Northeast Pacific marine heatwaves and Central Pacific El Niño events. *Geophysical Research Letters*, 49, e2021GL097350. https://doi. org/10.1029/2021GL097350

Received 7 DEC 2021 Accepted 23 FEB 2022

#### **Author Contributions:**

Conceptualization: A. Capotondi, M. Newman, E. Di Lorenzo Formal analysis: A. Capotondi Funding acquisition: A. Capotondi Methodology: A. Capotondi, M. Newman Software: A. Capotondi, M. Newman, T. Xu Validation: A. Capotondi, M. Newman

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# An Optimal Precursor of Northeast Pacific Marine Heatwaves and Central Pacific El Niño Events

A. Capotondi<sup>1,2</sup> , M. Newman<sup>1,2</sup> , T. Xu<sup>2</sup> , and E. Di Lorenzo<sup>3</sup>

<sup>1</sup>Cooperative Institute for Research in Environmental Sciences, University of Colorado, Boulder, CO, USA, <sup>2</sup>NOAA Physical Sciences Laboratory, Boulder, CO, USA, <sup>3</sup>Georgia Institute of Technology, Atlanta, GA, USA

**Abstract** The intensity of Northeast Pacific marine heatwaves (MHWs) has been related to local stochastic atmospheric forcing with limited predictability, but their evolution and persistence may be controlled by large-scale climate influences. A Linear Inverse Model containing both sea surface temperature and sea surface height (SSH) anomalies is used to identify the "optimal" conditions for observed Northeast Pacific MHW events that developed two-to-four seasons later. These optimal initial conditions include SSH anomalies that are responsible for most of the MHW growth, suggesting the relevance of subsurface ocean dynamics. Moreover, Northeast Pacific MHW growth occurs as part of a basin-scale dynamical mode that links the North Pacific to central equatorial Pacific El Niño events, whose subsequent development may lengthen MHW duration.

**Plain Language Summary** The Northeast Pacific Ocean has experienced episodes of intense and persistent abnormally warm conditions, also known as marine heatwaves (MHWs), with devastating ecological impacts. Being able to predict these extreme events a few seasons in advance is therefore very important, but has proven elusive. Understanding why marine heatwaves differ from one event to the next, especially in terms of their strength and how long they last, is limited due to our inability to predict aspects of daily weather that help drive these events. However, whether a new marine heatwave might occur, and if so how it might then evolve, may be partly understood by relating it to climate conditions extending throughout the Pacific. In this study, we use an empirical method to construct a model capturing how such climate conditions may change over time, and therefore whether a marine heatwave could be anticipated. Our model finds specific ocean surface conditions appearing to lead to the development of the Northeast Pacific marine heatwaves 6–12 months later. These conditions are related to a unique pattern of sea surface anomalies spanning the North Pacific and the central tropical Pacific, which also links extra-tropical marine heatwaves with subsequent tropical central Pacific warming.

# 1. Introduction

Starting in the Winter of 2013/14, the Northeast Pacific experienced extreme sea surface temperatures (SSTs), first developing in the center of the Gulf of Alaska (Bond et al., 2015) and subsequently extending along the US West Coast. While the resulting marine heatwave (MHW) underwent substantial variation in its intensity and spatial pattern, the entire event lasted over 2 years, ending in mid-2016 (Amaya et al., 2016).

The ecological consequences of this MHW were profound, including dramatic declines in primary productivity (Whitney, 2015), geographical shifts in marine species (Cavole et al., 2016; Peterson et al., 2016; Piatt et al., 2020; Sanford et al., 2019), massive die-offs of mammals and sea birds (Cavole et al., 2016), unusual whale entanglements and mortality (Santora et al., 2020), and prolonged harmful algal blooms (McCabe et al., 2016) resulting in closures of shellfish fisheries in Washington, Oregon and California. Thus, the ability to anticipate such extreme events would be extremely important for planning and decision-making by fisheries managers and other stakeholders. However, predicting the 2013/16 event has proved challenging (Hu et al., 2017; Jacox et al., 2019).

Some studies (Di Lorenzo & Mantua, 2016; Holbrook et al., 2019, 2020) have highlighted possible connections between Northeast Pacific MHWs and Pacific modes of climate variability, in particular El Niño Southern Oscillation (ENSO) in the tropical Pacific. Indeed, El Niño events are usually associated with warm conditions along the US West Coast (e.g., Capotondi, Jacox, et al., 2019, and references therein) and provide an important source of predictability for SST variability there (Jacox et al., 2017, 2020). However, ENSO events exhibit diverse spatial patterns (Capotondi et al., 2015; Capotondi, Wittenberg, et al., 2020) with potentially diverse impacts on Northeast Pacific SSTs (Shin & Newman, 2021). In particular, temperatures along the US West Coast appear to



Visualization: A. Capotondi, M. Newman Writing – original draft: A. Capotondi Writing – review & editing: M. Newman, T. Xu, E. Di Lorenzo be more sensitive to equatorial SST anomalies (SSTAs) in the central-western Pacific (Capotondi, Sardeshmukh, et al., 2019), highlighting the importance of accounting for this diversity of ENSO expressions when examining the ENSO-MHW connection.

Others (Amaya et al., 2021; Bond et al., 2015) have emphasized the role of persistent local atmospheric forcing, together with enhanced upper-ocean stratification and reduced mixed-layer depth as causes of the extreme intensity and persistence of Northeast Pacific MHWs. Recent results (Xu et al., 2021; XNCD hereafter) show that the extreme warm conditions that have recently occurred in the Northeast Pacific are part of a continuum of events with a broad range of intensities/durations (Scannel et al., 2016; XNCD), and with similar temporal evolutions. The duration of these events appears to be controlled by the timing of ENSO development in the equatorial Pacific, while the intensity is primarily influenced by local atmospheric forcing.

Similar evolution of these events suggests that large-scale climate conditions may help drive the development of Northeast Pacific MHWs. In this study, we thus build on XNCD to: (a) Ascertain that such large-scale conditions do exist and can provide some degree of MHW predictability; (b) Elucidate their dynamics, and the relative role of North and tropical Pacific influences; and (c) Examine the role of local and remote oceanic processes in the Northeast Pacific MHW onset.

The paper is organized as follows: In Section 2 we describe the data and methodology used, while in Section 3 we examine the composite evolution of northeast Pacific MHWs, and determine the large-scale precursors of MHWs. Conclusions are drawn in Section 4.

# 2. The Linear Inverse Model

The time evolution of a climate state x may often be approximated by the stochastically forced linear dynamical system (Penland & Sardeshmukh, 1995):

$$dx/dt = Lx + \xi \tag{1}$$

where L is a dynamical operator encapsulating the linear dynamics of the system, including an implicit linear parameterization of "fast" (decorrelation timescales much shorter than the dominant dynamical timescales) nonlinear processes, and  $\xi$  represents a stochastic white noise forcing that may be spatially coherent. For a given initial state x(t), the most likely state at a later time  $t + \tau$  is:

$$\mathbf{x}(t+\tau) = \mathbf{G}(\tau)\mathbf{x}(t) = \exp(L\tau)\mathbf{x}(t), \tag{2}$$

where we define the propagator operator  $G(\tau) = \exp(L\tau)$ . Determining Equation 1 from observed covariances results in a Linear Inverse Model (LIM; Penland & Sardeshmukh, 1995). LIMs are typically low-order models that are constructed from the leading Principal Component time series of a reduced empirical orthogonal function (EOF) space, but with seasonal forecast skill comparable to state-of-the-art global climate models (Shin & Newman, 2021). Further details concerning the LIM and its construction may be found in Penland and Sardeshmukh (1995), Newman et al. (2011), Capotondi and Sardeshmukh (2017), and XNCD.

In general, L is stable, so perturbations do not grow exponentially without bound. However, asymmetries in L (e.g., due to geographical variation of the climatological time-mean state and/or the interactions between variables) mean that its eigenvectors are generally not orthogonal. Under certain circumstances, this allows anomalies to undergo amplification over finite time intervals when different eigenvectors with similar spatial patterns, but different temporal scales, evolve from destructive to constructive interference (e.g., Farrell, 1988; Penland & Sardeshmukh, 1995). The eigenmode combinations that yield such transient growth can be determined through an analysis of the propagator matrix for the chosen time interval. This amplification can be measured over the whole domain (based on the so-called L2 norm; Text S1 and S2 in Supporting Information S1) or for specific patterns of interest. Importantly for our problem, we can use the LIM methodology to objectively identify initial conditions that optimally grow into a MHW pattern, referred to here as the "MHW-norm" (Text S1 in Supporting Information S1). This approach is applied for different "optimization times"  $\tau$ , that is, different growth periods, thus highlighting MHWs of different durations.

For our LIM, the variability over the tropical and North Pacific region  $(30^{\circ}\text{S}-70^{\circ}\text{N}, 100^{\circ}-290^{\circ}\text{E})$  is described by a state vector  $\mathbf{x} = [\text{SSTA SSHA}]$ , which contains monthly anomalies of SST and sea surface height (SSH), where





**Figure 1.** Marine heatwave (MHW) composites. (a) "NEPac index," computed as the average SSTA in the box displayed in b (middle), and subjected to a 3-month running average. Vertical light blue lines mark the times (month/year: 2/63, 9/86, 9/89, 1/91, 10/93, 6/04, 10/13) when the SSTAs first exceed the 1 standard deviation threshold, shown by the dotted orange line. (b) Composites of SSTAs for events exceeding 1 standard deviation of the NEPac index for at least 5 months, at lags -12, -6, 0, 6, and 12 months. The box in the panel at lag 0 is the region used to compute the NEPac index, while the meridional dashed lines indicate the band used for the Hovmöller diagrams in Figure 3. Thin black zonal and meridional lines mark the equator and the dateline, respectively. Hatching indicates values larger (lower) than the 95th (5th) percentiles of 500 randomly selected composites. (c) Same as in (b), but for SSH. (d) Composite of SSTAs at lag 0, zeroed south of 25°N, used to create the MHW norm. (e) Representation of (d) in the reduced SST space (20 EOFs) used to construct the LIM; that is, the "MHW-norm".

SSH provides a measure of upper ocean heat content and system's memory. Thus, our approach differs from that of XNCD, which only included SSTAs in their LIM's state vector. Including SSH in the LIM improves its ability to capture the evolution and diversity of ENSO events (Capotondi & Sardeshmukh, 2015; Newman et al., 2011), and might be expected to affect the development and timing of MHWs. Other specific details of the LIM used for this paper are in the supplemental material (Text S1 in Supporting Information S1).

Monthly SST and SSH data were obtained from the European Centre for Medium-Range Weather Forecasting (ECMWF) Ocean Reanalysis System 4 (ORAS4; Balmaseda et al., 2013), available to us (at the time of our analysis) from January 1958 to December 2015. To examine the large-scale atmospheric fields associated with the development and evolution of the optimal precursors' patterns we used sea level pressure (SLP) from the National Center for Environmental Prediction (NCEP)/National Center for Atmospheric Research (NCAR) reanalysis (Kalnay et al., 1996).

# 3. Results

#### 3.1. Constructing the Northeast Pacific MHW-Norm

Following XNCD's approach, we first construct a Northeast Pacific index ("NEPac"; Figure 1a) from 3-month running mean SSTAs averaged within the region 35°–55°N, 160°–130°W (box in Figure 1b, lag 0). As in previous studies of persistent climate anomalies (e.g., Breeden et al., 2020; XNCD), we define a MHW event as occurring when the NEPac index first exceeds a given amplitude and then remains above it for a given time period. Using XNCD's "intensity" and "duration" thresholds, one standard deviation and 5 months, respectively,

we find seven separate events. Note that XNCD found a few additional events for the same criteria due to the use of a longer time period and a different SST data set. The "onset" time of each event, when the index first exceeds the amplitude threshold, is marked by a light-blue vertical line in Figure 1a. The composite evolution of the SSTAs for these events, from 12 months prior to onset time (lag 0) to 12 months afterward, shows weak positive anomalies in the Gulf of Alaska and in the subtropical North Pacific at lag -12 months, while negative anomalies are seen in the eastern-central equatorial Pacific (Figure 1b). The initial anomalies in the Gulf of Alaska grow in intensity over the subsequent months, as an El Niño develops in the central equatorial Pacific (CP El Niño). A similar evolution is obtained when the 2013–2016 event is excluded from the composite (not shown). The composite MHW evolution also shows a pronounced SSH signature (Figure 1c), which was not examined in XNCD. Anomalies in the tropical Pacific underscore the El Niño development with the initial negative equatorial anomalies (shallower thermocline) slowly being replaced by positive anomalies, consistent with off-equatorial downwelling Rossby waves (positive SSH anomalies in the off-equatorial tropics) propagating to the western Pacific, continuing along the boundary to the equator as coastal Kelvin waves, and leading, 12 months after MHW onset, to the east-west SSH dipole typical of mature ENSO phases (e.g., Capotondi, Deser, et al., 2020; Meinen & McPhaden, 2000). Note that XNCD found that more extreme threshold criteria led to a higher amplitude composite and also a reduced sample size but did not qualitatively impact their analysis.

Finally, to construct the MHW norm, we create a "MHW onset composite" by retaining only anomalies north of 25°N in the lag 0 SST composite (Figure 1d). This pattern is then projected on the reduced EOF space of the LIM, resulting in the MHW-norm pattern ( $r_{SST}$  in Text S1 of Supporting Information S1) shown in Figure 1e. SSH anomalies (SSHAs) associated with the initial MHW composite have a very small signature north of 25°N when represented in the reduced EOF space, and therefore have not been considered in the definition of the MHW-norm.

#### 3.2. Optimal Structures Under the MHW-Norm

The two leading optimal structures maximizing SST and SSH growth over the full domain, according to the L2 norm, correspond to canonical and Central Pacific (CP) ENSO events (Text S2, Figures S1, and S2 in Supporting Information S1). We then isolate the specific initial condition  $x_{MHW}(0)$  maximizing Northeast Pacific MHW growth using the pattern in Figure 1e as our target (the "MHW-norm," Text S1 in Supporting Information S1). To quantify MHW anomaly growth, rather than overall anomaly growth, we first define MHW amplitude as the projection of an anomaly onto the MHW-norm. MHW growth is then defined as the difference between the MHW amplitudes for an evolving anomaly at its optimal (initial) and maximum (evolved) states. This varies for each optimization time interval  $\tau$ , as shown in Figure 2a, with a rapid increase at short optimization times, peak potential growth at around 7 months, and then a slow decrease at longer optimization times. Longer  $\tau$  values allow more time for amplification from a unit initial condition, but growth cannot occur indefinitely. The optimization time corresponding to the peak growth thus represents the interval over which the greatest possible growth can occur, with longer time intervals experiencing less net growth *over that interval*.

The SST and SSH anomalies for the MHW  $\tau = 9$  months optimal (that is, initial conditions optimizing MHW growth over a 9 months interval) are shown in Figures 2b and 2c, respectively, along with their subsequent evolution beneath. The initial patterns and evolutions obtained at optimization times between 6 and 12 months (where growth is largest) are very similar (e.g., Figure S3 in Supporting Information S1). Initially (t = 0), weak SSTAs are seen along the Gulf of Alaska rim, in the northern subtropics and along the equator, while larger SSH signals are seen, notably in the off-equatorial northern tropics (likely associated with Rossby wave activity) and along the Kuroshio Extension. Weak positive SSH anomalies are also present in the Northeast Pacific. As time progresses, large SSTAs develop in the center of the Gulf of Alaska, concurrent with the development of the Pacific Meridional Mode (PMM; Chiang & Vimont, 2004) in the subtropics (positive SSTAs extending southwestward from the coast of California toward the equatorial Pacific) and CP El Niño conditions along the equator. SSHAs also evolve into conditions typical of CP events (Capotondi, Wittenberg, et al., 2020): larger heat content (positive SSH) in the central-eastern equatorial Pacific and reduced heat content (negative SSH) in the far western Pacific.

Regression of SLP on the combined SST/SSH optimal time series indicates initial conditions characterized by the dipole pattern typical of the North Pacific Oscillation (NPO; Linkin & Nigam, 2008; Rogers, 1981), as previously noted for individual observed MHW events (Amaya et al., 2020; Bond et al., 2015), which slowly evolves into a deepened Aleutian Low, as El Niño develops along the equator (Di Lorenzo & Mantua, 2016).





**Figure 2.** Marine heatwave (MHW) optimal. (a) Maximum potential MHW growth as a function of optimization time. The MHW optimal at each optimization time is evolved in time and projected on the MHW norm. Growth is defined as the change in MHW amplitude from initial to maximum value. (b) Evolution of the SSTA component of the MHW optimal for optimization time  $\tau = 9$  months, from the initial (t = 0) to the final (t = 9) time. Black contours show the regression of SLP on the optimal time series at each time level. Contour interval is 0.15, negative values dashed. (c) Evolution of the SSHA component of the MHW optimal for  $\tau = 9$  months.





**Figure 3.** Evolution of marine heatwave (MHW) optimals determined from different optimization times. (a–c) MHW amplitude as a function of time for different optimization time intervals of either (a)  $\tau = 6$ , (b)  $\tau = 9$ , or (c)  $\tau = 12$  months (d–f) The evolution of the optimals averaged zonally over the 160°-130°W longitude strip (dashed meridional lines in Figure 1b) as a function of time (*x*-axis) and latitude (*y*-axis), shown for (d)  $\tau = 6$ , (e)  $\tau = 9$ , and (f)  $\tau = 12$  months. Shading indicates SSTA (C.I. 0.06), while contours indicate SSHA (C.I. 0.015, negative values dashed). Units are arbitrary, but normalization is consistent among panels. (g–i) The projection coefficients of observed anomalies on the corresponding optimal patterns, versus the observed NEPac index (g) 6, (h) 9, or (i) 12 months later. The dashed lines indicate the linear fit using a least squares approach. Correlations (insets) are significant at the 99% level based on an effective number of degrees of freedom (Bretherton et al., 1999). Black dots indicate the NEPac index at the onset times of the seven identified MHWs (*y*-axis) versus the projection coefficients  $\tau$  months earlier (*x*-axis).

Key characteristics of the MHW-Optimal are compared in Figure 3 for different optimization time intervals ( $\tau = 6, 9$ , or 12 months). In each case, the MHW amplitude grows for a period of approximately  $\tau$  months, by which time it reaches its maximum value (Figures 3a–3c). Exploiting the linearity of the LIM, we examine the relative contribution of the initial SST and SSH signals to the MHW growth by zeroing either the SST or SSH components of the initial MHW optimal. While the absence of the initial SSTAs only slightly reduces MHW growth (red lines in Figures 3a–3c), removal of the initial SSH signal notably hinders growth (blue lines). Inspection of the initial evolution of the SSTAs in the two cases (SSTA = 0 or SSHA = 0 at t = 0) shows that the presence of initial SSHAs allows the generation of SSTAs already at t = 1, even if they were set to zero at t = 0 (Figure S4 in Supporting Information S1). The time evolution of SST and SSH anomalies averaged within an eastern Pacific longitude band (dotted meridional lines in Figure 1b) shows anomaly growth both in the northeast Pacific MHW region and, slightly lagged, in the tropics (middle row of Figure 3). For longer optimization times, MHW anomalies have reduced amplitude but increased duration (cf., Figures 3d and 3f), with year-long events having a double maximum similar to the 2014–2016 event. Enhanced persistence and an eastward shift of MHW anomalies is associated with longer-lasting warm conditions in the equatorial Pacific, as found by XNCD.

The potential growth captured by the MHW optimals is observed to occur. Comparison of the projection of the SST/SSH fields on the optimal patterns at time t and the observed NEPac index at time  $t + \tau$  (Figure 3, bottom panels) shows fairly linear relationships, with statistically significant correlation coefficients 0.61, 0.54,

and 0.50 for optimization times of 6, 9, and 12 months, respectively. Scatter is consistent with the presence of local stochastic atmospheric forcing of the MHW events, which may impact individual event amplitude (Bond et al., 2015, XNCD).

#### 3.3. Large-Scale Dynamics Linking MHW and CP El Niño Events

The connection between Northeast Pacific and equatorial CP warming raises the question of how the North Pacific and the tropics are linked. To further investigate this issue, we examine the eigenmodes of the LIM dynamical operator L to identify which most contribute to MHW optimals and their growth (Text S3, Figures S5, and S6 in Supporting Information S1). The eigenmodes of L (Figure S5 in Supporting Information S1) are all stable and can be either stationary (real) or propagating (complex conjugate pairs). Since we are interested in events lasting at least two seasons, we focus here on eigenmodes with e-folding times greater than 5–6 months, although other more strongly damped eigenmodes (shorter e-folding times) may contribute to initial anomaly growth and evolution. We identify three such eigenmodes (Figure S6 in Supporting Information S1) that correspond to different flavors of ENSO: an "NP-CP" eigenmode with SSTAs spanning both the North Pacific and the central equatorial Pacific, and two eigenmodes whose anomalies are primarily confined within the tropics. Of these, the NP-CP eigenmode appears to play the key role in MHW growth, with its most energetic phase (Figure S6b in Supporting Information S1) dominating the mature phase of the MHW optimal (e.g., Figure 2b, t = 9) for both SST and SSH. Since for any nonnormal linear dynamical system, the optimal initial condition for asymptotic growth of an eigenmode is not itself, but rather its adjoint (i.e., the corresponding eigenvector of  $L^{\dagger}$ , see Text S3 in Supporting Information S1 for details; Farrell, 1988), we compared evolution starting from an MHW optimal to that starting from an initial state corresponding to the adjoint of the NP-CP eigenmode (Figure 4). The pattern of the initial adjoint, with appropriate normalization and phase rotation, is quite similar to the MHW optimal (top row of Figure 4), and the resulting evolution of both is also strikingly similar. That is, the initial condition leading to optimal development of Northeast Pacific MHWs is essentially the initial condition leading to optimal development of the NP-CP eigenmode. Similar comparisons are found at different optimization times (Figure S7 in Supporting Information S1), although the correspondence between adjoint and optimal is reduced at shorter  $\tau$ , since the influence of other faster decaying eigenmodes is greater then.

# 4. Conclusions

A LIM is used to objectively identify large-scale precursors conducive to the dynamical development of Northeast Pacific MHWs on seasonal time scales. Notably, we find that Northeast Pacific MHWs and subsequent CP EI Niño events may be dynamically linked through the optimal evolution of a single dynamical mode, the "NP-CP" eigenmode, whose most energetic phase includes anomalies in the Northeast Pacific, the northern subtropics and the central tropical Pacific. The spatial structure of the mode is consistent with the two-way teleconnection between the North Pacific and the tropics identified in previous studies as a source of extended climate memory and decadal variability (Di Lorenzo et al., 2015; Zhao & Di Lorenzo, 2020). The evolution of the mode is also consistent with the development of those El Niño events that impact US West Coast SSTs (Capotondi, Sardeshmukh, et al., 2019). This mode was also identified as a component of the Pacific Decadal Oscillation (Newman et al., 2016). Moreover, the pronounced similarity of the optimal precursor of Northeast Pacific MHWs to the adjoint of the NP-CP eigenmode is particularly striking. Since the "stochastic optimal" noise forcing of an eigenmode is related to its adjoint (Moore & Kleeman, 1999), this similarity suggests that some MHW and CP El Niño events may be first triggered by the same initial noise event, with subsequent deterministic development from MHW through CP El Niño due to dynamical processes that are not yet fully understood.

The LIM used in this study does not explicitly account for seasonal variation of the deterministic dynamics, which we know are important for both ENSO evolution, and for the processes underpinning MHW growth. The onset times (September-to-February) of six of the seven MHWs identified in our record (Figure 1a) are consistent with the seasonal development of the Pacific Meridional Mode, and the subsequent evolution of CP El Niño events, as discussed in a recent study that used a seasonally varying LIM (Vimont et al., 2022), suggesting that some seasonal aspects of MHW and CP ENSO evolution may be implicitly captured by our LIM. A seasonally varying LIM (Shin et al., 2021) will be considered in future studies.





**Figure 4.** Large-scale dynamics linking marine heatwaves (MHWs) and CP El Niño. (a) Evolution starting from the adjoint of the "NP-CP" eigenmode. The pattern of the adjoint at time t = 0 corresponds to the phase of its maximum projection (0.79) on the  $\tau = 9$  months MHW optimal at t = 0. Shading indicates sea surface temperature, while contours are used for sea surface height. Contour interval is 0.02, with negative values dashed. (b) Evolution starting from the MHW optimal, displayed for comparison with that from the adjoint. (c) Difference between the evolutions starting from the adjoint and the MHW optimal.

An intriguing result of our analysis is that initial SSH anomalies are essential for MHW growth, and can enable that growth even in the absence of initial SST anomalies. While this result suggests a potentially important role for oceanic processes in the development of Northeast Pacific MHWs, the exact nature of these processes is unclear at the moment, and will be investigated in future work.

# **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

#### **Data Availability Statement**

The output of the ORAS4 ocean reanalysis was obtained from https://www.cen.uni-hamburg.de/en/icdc/data/ ocean/easy-init-ocean/ecmwf-ocean-reanalysis-system-4-oras4.html website. The NCEP/NCAR SLP and geopotential height data are available from the NOAA/PSL website (https://psl.noaa.gov/data/gridded/data.ncep. reanalysis.derived.html).

#### References

- Amaya, D. J., Alexander, A. A., Capotondi, A., Deser, C., Karnauskas, K. B., Miller, A. J., & Mantua, N. J. (2021). Are long-term changes in mixed layer depth influencing North Pacific marine heatwaves? *Bulletin of the American Meteorological Society*, 102(1), S59–S66. https:// doi.org/10.1175/bams-d-20-0144.1
- Amaya, D. J., Bond, N., Miller, A., & DeFlorio, M. (2016). The evolution and known atmospheric forcing mechanisms behind the 2013–2015 North Pacific warm anomalies. CLIVAR's Variations Newsletter. Spring, 1–6.
- Amaya, D. J., Miller, A. J., Xie, S.-P., & Kosaka, Y. (2020). Physical drivers of the summer 2019 North Pacific marine heatwave. Nature Communications, 11(1), 1903. https://doi.org/10.1038/s41467-020-15820-w
- Balmaseda, M. A., Mogensen, K., & Weaver, A. T. (2013). Evaluation of the ECMWF ocean reanalysis system ORAS4. Quarterly Journal of the Royal Meteorological Society, 139(674), 1132–1161. https://doi.org/10.1002/qj.2063
- Bond, N. A., Cronin, M. F., Freeland, H., & Mantua, N. (2015). Causes and impacts of the 2014 warm anomaly in the NE Pacific. *Geophysical Research Letters*, 42(9), 3414–3420. https://doi.org/10.1002/2015GL063306
- Breeden, M. L., Hoover, B. T., Newman, M., & Vimont, D. J. (2020). Optimal North Pacific blocking precursors and their deterministic sub-seasonal evolution during boreal winter. *Monthly Weather Review*, 148(2), 739–761. https://doi.org/10.1175/mwr-d-19-0273.1
- Bretherton, C. S., Widmann, M., Dymnikov, V. P., Wallace, J. M., & Bladé, I. (1999). The effective number of spatial degrees of freedom of a time-varying field. *Journal of Climate*, 12(7), 1990–2009. https://doi.org/10.1175/1520-0442(1999)012<1990:tenosd>2.0.co;2
- Capotondi, A., Deser, C., Phillips, A. S., Okumura, Y., & Larson, S. M. (2020). ENSO and Pacific decadal variability in the Community Earth System Model version 2. Journal of Advances in Modeling Earth Systems, 12. https://doi.org/10.1029/2019MS002022
- Capotondi, A., Jacox, M., Bowler, C., Kavanaugh, M., Lehodey, P., Barrie, D., et al. (2019). Observational needs supporting marine ecosystems modeling and forecasting: From the global oceans to the regional and coastal systems. *Frontiers in Marine Science*, 6. https://doi.org/10.3389/ fmars.2019.00623
- Capotondi, A., & Sardeshmukh, P. D. (2015). Optimal precursors of different types of ENSO events. *Geophysical Research Letters*, 42(22), 9952–9960. https://doi.org/10.1002/2015GL066171
- Capotondi, A., & Sardeshmukh, P. D. (2017). Is El Niño really changing? *Geophysical Research Letters*, 44(16), 8548–8556. https://doi.org/10.1002/2017GL074515
- Capotondi, A., Sardeshmukh, P. D., Di Lorenzo, E., Subramanian, A. C., & Miller, A. J. (2019). Predictability of US West Coast ocean temperatures is not solely due to ENSO. Scientific Reports, 9(1), 10993. https://doi.org/10.1038/s41598-019-47400-4
- Capotondi, A., Wittenberg, A. T., Kug, J.-S., Takahashi, K., & McPhaden, M. (2020). ENSO diversity. In A. Santoso, W. Cai, & M. McPhaden (Eds.), *El Niño Southern Oscillation in a changing climate* (pp. 65–86). American Geophysical Union (AGU). https://doi.org/10.1002/9781119548164.ch4
- Capotondi, A., Wittenberg, A. T., Newman, M., Di Lorenzo, E., Yu, J. Y., Braconnot, P., et al. (2015). Understanding ENSO diversity. Bulletin of the American Meteorological Society, 96(6), 921–938. https://doi.org/10.1175/bams-d-13-00117.1
- Cavole, L. M., Demko, A. M., Diner, R. E., Giddings, A., Koester, I., Pagniello, C. M. L. S., et al. (2016). Biological impacts of the 2013–2015 warm-water anomaly in the Northeast Pacific. *Oceanography*, 29, 273–285.
- Chiang, J. C. H., & Vimont, D. J. (2004). Analogous Pacific and Atlantic meridional modes of tropical atmosphere–ocean variability. *Journal of Climate*, *17*(21), 4143–4158. https://doi.org/10.1175/jcli4953.1
- Di Lorenzo, E., Liguori, G., Schneider, N., Furtado, J. C., Anderson, B. T., & Alexander, M. A. (2015). ENSO and meridional modes: A null hypothesis for Pacific climate variability. *Geophysical Research Letters*, 42(21), 9440–9448. https://doi.org/10.1002/2015gl066281
- Di Lorenzo, E., & Mantua, N. (2016). Multi-year persistence of the 2014/15 North Pacific marine heatwave. Nature Climate Change, 6(11), 1042–1048. https://doi.org/10.1038/nclimate3082
- Farrell, B. (1988). Optimal excitation of neutral Rossby waves. Journal of the Atmospheric Sciences, 45(2), 163–172. https://doi.org/10.1175/15 20-0469(1988)045<0163:oeonrw>2.0.co;2
- Holbrook, N. J., Scannell, H. A., Sen Gupta, A., Benthuysen, J. A., Feng, M., Oliver, E. C. J., et al. (2019). A global assessment of marine heatwaves and their drivers. *Nature Communications*, 10(1), 13. https://doi.org/10.1038/s41467-019-10206-z
- Holbrook, N. J., Sen Gupta, A., Gupta, A. S., Oliver, E. C. J., Eric, C. J., Oliver, A. J., et al. (2020). Keeping pace with marine heatwaves. Nature Reviews Earth & Environment, 1(9), 482–493. https://doi.org/10.1038/s43017-020-0068-4
- Hu, Z. Z., Kumar, A., Jha, B., Zhu, J. S., & Huang, B. H. (2017). Persistence and predictions of the remarkable warm anomaly in the northeastern Pacific Ocean during 2014–16. Journal of Climate, 30(2), 689–702. https://doi.org/10.1175/jcli-d-16-0348.1
- Jacox, M. G., Alexander, M. A., Siedlecki, S., Chen, K., Kwon, Y. O., Brodie, S., et al. (2020). Seasonal-to-interannual prediction of North American coastal marine ecosystems: Forecast methods, mechanisms of predictability, and priority developments. *Progress in Oceanography*, 183, 102307. https://doi.org/10.1016/j.pocean.2020.102307
- Jacox, M. G., Alexander, M. A., Stock, C. A., & Hervieux, G. (2017). On the skill of seasonal sea surface temperature forecasts in the California Current System and its connection with ENSO variability. *Climate Dynamics*, 53(12), 7519–7533. https://doi.org/10.1007/s00382-017-3608-y
- Jacox, M. G., Tommasi, D., Alexander, M. A., Hervieux, G., & Stock, C. A. (2019). Predicting the evolution of the 2014–2016 California Current System marine heatwave from an ensemble of coupled global climate forecasts. *Frontiers in Marine Science*, 6, 13. https://doi.org/10.3389/ fmars.2019.00497
- Kalnay, E., Kanamitsu, M., Kistler, R., Collins, W., Deaven, D., Gandin, L., et al. (1996). The NCEP/NCAR 40-year reanalysis project. *Bulletin of the American Meteorological Society*, 77(3), 437–471. https://doi.org/10.1175/1520-0477(1996)077<0437:tnyrp>2.0.co;2
- Linkin, M. E., & Nigam, S. (2008). The North Pacific Oscillation-west Pacific teleconnection pattern: Mature-phase structure and winter impacts. Journal of Climate, 21(9), 1979–1997. https://doi.org/10.1175/2007jcli2048.1
- McCabe, R. M., Hickey, B. M., Kudela, R. M., Lefebvre, K. A., Adams, N. G., Bill, B. D., et al. (2016). An unprecedented coastwide toxic algal bloom linked to anomalous ocean conditions. *Geophysical Research Letters*, 43(19), 10366–10376. https://doi.org/10.1002/2016GL070023

#### Acknowledgments

A. Capotondi was supported by the Competitive Climate Research Portfolio, and by the NOAA Climate Program Office Modeling, Analysis, Prediction and Projection (MAPP) Program. M. Newman, T. Xu, and E. Di Lorenzo were supported by DOE grant #0000238382. The authors thank Drs. Nate Mantua and Alex Sen Gupta for their excellent comments and suggestions, which have greatly improved the manuscript.

- Meinen, C. S., & McPhaden, M. J. (2000). Observations of warm water volume changes in the equatorial Pacific and their relationship to El Niño and La Niña. *Journal of Climate*, *13*(20), 3551–3559. https://doi.org/10.1175/1520-0442(2000)013<3551:oowwvc>2.0.co;2
- Moore, A. M., & Kleeman, R. (1999). The nonnormal nature of El Niño and intraseasonal variability. Journal of Climate, 12(10), 2965–2982. https://doi.org/10.1175/1520-0442(1999)012<2965:tnnoen>2.0.co;2
- Newman, M., Alexander, M. A., Ault, T. R., Cobb, K. M., Deser, C., Di Lorenzo, E., et al. (2016). The Pacific decadal oscillation, revisited. Journal of Climate, 29(12), 4399–4427. https://doi.org/10.1175/JCLI-D-15-0508.1
- Newman, M., Shin, S. I., & Alexander, M. A. (2011). Natural variation in ENSO flavors. Geophysical Research Letters, 38(14), L14705. https:// doi.org/10.1029/2011GL047658
- Penland, C., & Sardeshmukh, P. D. (1995). The optimal-growth of tropical sea-surface temperature anomalies. *Journal of Climate*, 8, 1999–2024. https://doi.org/10.1175/1520-0442(1995)008<1999:togots>2.0.co;2

Peterson, W., Bond, N., & Robert, M. (2016). The Blob (part three): Going, going, gone? PICES Press, 24, 46-48.

- Piatt, J. F., Parrish, J. K., Renner, H. M., Schoen, S. K., Jones, T. T., Arimitsu, M. L., et al. (2020). Extreme mortality and reproductive failure of common murres resulting from the northeast Pacific marine heatwave of 2014–2016. PLoS One, 15(1), e0226087. https://doi.org/10.1371/ journal.pone.0226087
- Rogers, J. C. (1981). The north Pacific oscillation. Journal of Climatology, 1, 39-57. https://doi.org/10.1002/joc.3370010106
- Sanford, E., Sones, J. L., García-Reyes, M., Goddard, J. H. R., & Largier, J. L. (2019). Widespread shifts in the coastal biota of northern California during the 2014–2016 marine heatwaves. *Scientific Reports*, 9(1), 4216. https://doi.org/10.1038/s41598-019-40784-3
- Santora, J. A., Mantua, N. J., Schroeder, I. D., Field, J. C., Hazen, E. L., Bograd, S. J., et al. (2020). Habitat compression and ecosystem shifts as potential links between marine heatwave and record whale entanglements. *Nature Communications*, 11(1), 536. https://doi.org/10.1038/ s41467-019-14215-w
- Scannel, H. A., Pershing, A. J., Alexander, M. A., Thomas, A. C., & Mills, K. E. (2016). Frequency of marine heatwaves in the North Atlantic and North Pacific since 1950. *Geophysical Research Letters*, 43(5), 2069–2076. https://doi.org/10.1002/2015gl067308
- Shin, S.-I., & Newman, M. (2021). Seasonal predictability of global and North American coastal sea surface temperature and height anomalies. Geophysical Research Letters, 48(10), e2020GL091886. https://doi.org/10.1029/2020GL091886
- Shin, S.-I., Sardeshmukh, P. D., Newman, M., Penland, C., & Alexander, M. A. (2021). Impact of annual cycle on ENSO variability and predictability. *Journal of Climate*, 34(1), 171–193. https://doi.org/10.1175/JCLI-D-20-0291.1
- Vimont, D. J., Newman, N., Battisti, D. S., & Shin, S.-I. (2022). The role of seasonality and the ENSO mode in central and east Pacific ENSO growth and evolution. *Journal of Climate*, 1–46. https://doi.org/10.1175/JCLI-D-21-0599.1
- Whitney, F. A. (2015). Anomalous winter winds decrease 2014 transition zone productivity in the NE Pacific. *Geophysical Research Letters*, 42(2), 428–431. https://doi.org/10.1002/2014g1062634
- Xu, T., Newman, M., Capotondi, A., & Di Lorenzo, E. (2021). The continuum of Northeast Pacific marine heatwaves and their relationship to the tropical Pacific. *Geophysical Research Letters*, 48(2), e2020GL090661. https://doi.org/10.1029/2020GL090661
- Zhao, Y., & Di Lorenzo, E. (2020). The impacts of extra-tropical ENSO precursors on tropical Pacific decadal-scale variability. *Scientific Reports*, 10(1), 3031. https://doi.org/10.1038/s41598-020-59253-3

# **References From the Supporting Information**

- Ashok, K., Behera, S. K., Rao, S. A., Weng, H., & Yamagata, T. (2007). El Niño Modoki and its possible teleconnection. Journal of Geophysical Research, 112(C11), C11007. https://doi.org/10.1029/2006JC003798
- Newman, M. (2007). Interannual to decadal predictability of tropical and North Pacific sea surface temperatures. *Journal of Climate*, 20(11), 2333–2356. https://doi.org/10.1175/jcli4165.1
- Newman, M., Sardeshmukh, P. D., & Penland, C. (2009). How important is air-sea coupling in ENSO and MJO evolution? *Journal of Climate*, 22(11), 2958–2977. https://doi.org/10.1175/2008jcli2659.1
- Zhang, H., Clement, A., & Di Nezio, P. P. (2014). The south Pacific meridional mode: A mechanism for ENSO-like variability. *Journal of Climate*, 27(2), 769–783. https://doi.org/10.1175/jcli-d-13-00082.1